Author's response to reviews

Title: Assessment of water, sanitation, and hygiene practice and associated factors in a Buruli ulcer-endemic district in Benin (West Africa)

Authors:

Roch Christian JOHNSON (rochjohnson@yahoo.fr)
Gratien BONI (bonigras@yahoo.fr)
Yves BAROGUI (yvesbaro@yahoo.fr)
Ghislain Emmanuel SOPH (ghislainsop@yahoo.fr)
Macaire HOUNDONOUGBO (hbricemacaire@gmail.com)
Didier AGOSSADOU (didieragossadou@yahoo.fr)
Gabriel DIEZ (gabriediez@anesvad.org)
Michel BOKO (bokomichel@gmail.com)

Version:3 Date:3 June 2015

Author's response to reviews:

Dear Editors,

We would like to submit the second revised version of the manuscript entitled “Assessment of water, sanitation, and hygiene practices and associated factors in a Buruli ulcer-endemic district in Benin (West Africa),” for consideration for publication in BMC Public Health.

We would like to thank the reviewers for comments made to improve the quality of the manuscript. Below the point by point answers to questions raised by the reviewers.

Thank you for considering this revised version.

Sincerely

Dr Roch Christian JOHNSON

Reviewer 1
Line 207-208 page 8: the sentence is rephrased according to the comments of the reviewer.
The figure 1 is deleted in the text
Line 486: Corrected as requested by the reviewer.
Reviewer 2
Line 129
We would like to thank the reviewer for the comment. The sentences have been corrected accordingly to the suggestion
Line 208
The sentence is rephrased as suggested by the reviewer
Line 211
Corrected.
Line 217
Corrected
Line 232
The sentence is rephrased according to the suggestion of the reviewer
The p-values are stated in the discussion as requested.